Author's response to reviews

Title: Retinal upregulation of inflammatory and proangiogenic markers in a model of neonatal diabetic rats fed on a high-fat-diet

Authors:

Jorge E Mancini (dr.jmancini@me.com)
Gustavo A Ortiz (g_a_ortiz@yahoo.com.ar)
Juan O Croxatto (juan.croxatto@gmail.com)
Juan E Gallo (jgallo06@gmail.com)

Version: 4 Date: 16 February 2013

Author's response to reviews: see over
Cover Letter

15 February, 2013

BMC Ophthalmology
Editorial Office

Re: MS 1438013038829097; Retinal upregulation of inflammatory and proangiogenic markers in a model of neonatal diabetic rats fed on a high-fat-diet.

Dear Sirs,

Please find enclosed the revised version of our manuscript including almost all the modifications requested by the reviewers. Besides, we have written in this letter point-by-point our answers to the reviewers.

Independently of the action that the journal takes we would like to thank BMC-Ophthalmology for having improved the quality of our manuscript.

Yours sincerely,

Juan E. Gallo, M.D., PhD